THE WAVELENGTH SPECTRUM SHIFT OF A CAVITY-DUMPED ARGON LASER-PUMPED RHODAMINE 6-G ORGANIC-DYE LASER

Gerald Warner Snyder

Library Naval Postgraduate School Monterey, California 93940

NAVAL POSTGRADUATE SCHOOL Monterey, California





THE WAVELENGTH SPECTRUM SHIFT OF A CAVITY-DUMPED ARGON LASER-PUMPED RHODAMINE 6-G ORGANIC-DYE LASER

Ъу

Gerald Warner Snyder

Thesis Advisor:

John P. Powers

T156412

September 1973

Approved for public release; distribution unlimited.



The Wavelength Spectrum Shift of a Cavity-Dumped Argon Laser-Pumped Rhodamine 6-G Organic-Dye Laser

Ъy

Gerald Warner Snyder Lieutenant Commander, United States Navy B. S., United States Naval Academy, 1964

Submitted in partial fulfillment of the requirements for the degree of

ELECTRICAL ENGINEER

from the NAVAL POSTGRADUATE SCHOOL September 1973

Thes 5 5 0 6 7 1 C.

Library Naval Postgraduate School Monterey, California 03840

ABSTRACT

The wavelength spectrum of a CW pumped Rhodamine 6-G organic dye laser is observed to shift approximately 150 Angstroms toward shorter wavelengths when pumped by 30 nsec pulses at a one megahertz repetition rate from a cavity-dumped Argon laser. Experimental evidence of the shift is presented, gain and rate equations are developed for a simplified dye laser model, and theoretical results are obtained for a computer simulation of the experiment. A comparison is made of the theoretical and experimental results and satisfactory agreement is obtained within the limits of the values of the parameters used and the assumptions made in formulating the model.

TABLE OF CONTENTS

I.	INTI	RODUCTION
II.	DYE	LASER THEORY 8
	Α.	ORGANIC DYE PROPERTIES 8
		1. Energy Level Description 8
		2. Dye Laser Process 8
	в.	DYE LASER MATHEMATICS11
		1. Dye Laser Gain Equation11
		2. Dye Laser Rate Equations15
III.	EXP:	ERIMENTAL PROCEDURES AND RESULTS
	Α.	EXPERIMENTAL SET UP
	в.	EXPERIMENTAL PROCEDURES
	С.	EXPERIMENTAL RESULTS
IV.	DYE	LASER MODEL EVALUATION
	Α.	LITERATURE SURVEY
	в.	GAIN AND RATE EQUATION PARAMETERS
	C.	GAIN AND RATE EQUATION SOLUTIONS
		1. Solution Method
		2. Theoretical Results
ν.	CON	CLUSIONS
	Α.	EXPERIMENTAL AND THEOREFICAL COMPARISON
	В.	REMARKS
APPENDIX A Computer Program Listing		
BIBLIOGRAPHY		
INITIAL DISTRIBUTION LIST		
FORM DD 1473		



ACKNOWLEDGEMENT

I am grateful for the help and support of Professor John Powers of the U.S. Naval Postgraduate School. I also want to thank Jim Jernigan of Code 6043, U.S. Naval Weapons Center, China Lake, California for the use of equipment and technical advice.

I. INTRODUCTION

Man's quest for knowledge, inherent curiosity, and need for a tunable coherent source of light have rapidly advanced the development of the organic dye laser since stimulated emission was observed from Trivalent Uranium (a four level laser system) by Sorokin and Stevenson [1] in 1960.

Brock, et al. [2] in 1961 proposed that organic compounds could be used as a laser medium, but stimulated emission from an organic material was not observed until 1962 when Morantz, White and Wright [3] observed it from benzophenone and napthalene imbedded in a glass matrix. This was followed in 1963 by the observation of stimulated processes from Eubenzoylacetonate by Lempicki and Samuelson [4].

Stimulated emission from an organic dye, however, was not accomplished until 1966 when it was observed by Sorokin and Lankard [5] from Chloroaluminum phthalocyanine dissolved in ethyl alcohol when pumped by a giant pulse ruby laser. Sorokin and Lankard [6] also reported in 1967 the first observed stimulated emission from flashlamp pumped Acridine Red, Rhodamine 6-G (the most common dye used in present day organic dye lasers), and Fluorescein. Schafer, Schmidt, and Volze [7] reported stimulated emission from several other organic dyes during this same time period while using the same basic procedure as Sorokin and Lankard [5], and they were the first to suggest that a dye laser operated on transitions from the excited singlet state to vibrational levels of the ground singlet state.

The feasibility of a continuous wave dye laser was discussed by Snavely [8] and was demonstrated by Snavely and Schafer [9] in 1969. CW operation was achieved in 1970 by Peterson, Tuccio and Snavely [10] and

Hercher and Pike [11] using an Argon ion laser as the pump source and Rhodamine 6-G as the organic dye.

The first CW dye lasers were only about one percent efficient. This has been raised to the present day level of thirty-five percent through the efforts of several groups of people including Hercher and Pike [12, 13], Kohn, Shank, Ippen and Dienes [14, 15], and Tuccio and Strome [16, 17].

The evolution of the dye laser has seen it constructed in many ways. The tunable distributed-feedback dye laser built by Shank, Bjorkholm and Kogelnik [18], the prism-dye laser constructed by Chandra, Takeuchi, and Hartmann [19], and the evanescent-field-pumped dye laser demonstrated by Ippen and Shank [20] are a few of the many sophisticated methods that have been used to obtain dye laser emission. The most common dye laser configuration, however, has the pump and dye laser cavities aligned with the beams coincident and the dye flowing perpendicular to the pump beam through a dye cell.

Several uses have been proposed for dye lasers. Bloom [21] has suggested its use for attenuation measurements on narrow-band absorption lines in molecular gases, and Sorokin, Lankard, Moruzzi and Hammond [22] have proposed using the dye laser for optical studies of rare-earth ions, photochemistry, and double-quantum absorption spectroscopy. The dye laser characteristics which make it exceptionally well suited to the above uses are its tunable range which can be as much as 1100 Angstroms from a single dye, and its narrow linewidth which is normally less than 0.5 Angstroms.

Uses for the dye laser which have already been published include infrared difference-frequency generation [23], megawatt tunable second harmonic and sum frequency generation [24], studies of the Sodium D

the second se

resonance lines by high resolution Spectroscopy [25], and detection of OH in the atmosphere [26]. Further uses of the dye laser are limited only by the interest and ingenuity of man.

In this work it has been observed that the output wavelength spectrum of a CW pumped Rhodamine 6-G organic dye laser shifts toward shorter wavelengths when pumped by short intense pulses from a cavity-dumped Argon laser. This shift to the green region of the visible wavelength spectrum is important in that the new spectrum can be matched to the wavelength response curve of a crystal to produce a tunable dye laser optical memory system. The new spectrum also provides the opportunity to investigate laser propagation in blue and green sea water over an appreciable wavelength range using a single dye.

The purpose of this thesis is to provide experimental results of the observed spectrum shift, to obtain a computer simulation of a dye laser model, and to compare experimental and theoretical results. Section II discusses dye laser theory based on energy level considerations and gives the development of the dye laser gain and rate equations for an assumed physical model. Section III presents experimental procedures and results and Section IV discusses the solution procedures and results of the theoretical model. Section V provides a comparison of experimental and theoretical results and conclusions. The computer program used to solve the dye laser model is listed in Appendix A.

II. DYE LASER THEORY

A. ORGANIC DYE PROPERTIES

1. Energy Level Description

A typical energy level diagram for an organic dye is shown in Figure 1. The reference level A is the electronic ground state of the molecular singlet state S_0 . S_1 and S_2 are the first and second excited singlet states, and T_1 and T_2 are the first and second excited triplet states. The small letters a and b indicate molecular vibrational energy levels and the primed letters indicate molecular rotational energy levels within a state. The typical separation of vibrational energy levels is 0.1 electron volts and that of rotational energy levels is 0.001 electron volts [27] so that each state may be viewed as a continuous band of energy.

2. Dye Laser Process

The dye laser process starts with the excitation of molecules from level A of S_0 to an upper vibrational or rotational level of S_1 . The excited molecules then decay very rapidly $(10^{-10} - 10^{-13} \text{ sec})$ [28] by nonradiative internal conversion to B.

The excited molecule in B has three options. It may decay spontaneously to a or a' (called fluorescence), it may undergo a stimulated transition to a or a', or it may travel via intersystem crossing to the lower level of T_1 .

Fluorescence depends on the natural lifetime of B. The fluorescence spectrum of an organic dyc is governed by the Franck-Condon principle [29] which states that preferred electron transitions are determined by the wave functions of the individual energy levels.



Monradiative decay-energy converted to heat

Figure 1: Organic dye energy levels.

Stimulated transition to a or a' can occur only if the excitation pulse is fast and intense enough to create a much larger than normal concentration of molecules in S_1 , called critical inversion, so that coherent emission may take place from the dye. The critical inversion as well as the emission wavelength are governed by losses in the dye laser system. Once the excited molecule reaches a or a' it decays nonradiatively to A.

Intersystem crossing is the least desirable transition for an excited molecule. The singlet state is characterized by opposite electron spins and the triplet state by parallel electron spins. As a result, according to the laws of quantum mechanics, the $S_1 - T_1$ transition is spin forbidden and is in fact approximately 10^{-6} [30] less probable than the $S_1 - S_0$ transition. However, the $S_1 - T_1$ transition does occur and is significant enough to have an important negative effect on the dye laser process.

The intersystem crossing rate time constant, k_{st}^{-1} , is generally much smaller than the T_1 state lifetime, T_t , therefore, state T_1 acts as a time dependent trap for dye molecules. This not only reduces the possible dye laser efficiency by removing dye molecules from the singlet system, but also, since the absorption spectrum of the $T_1 - T_2$ transition of an organic dye usually overlaps the dye fluorescence spectrum, another loss mechanism is added to the dye laser system. $S_1 - S_2$ absorption is possible but it is usually neglected because the fraction of molecules in S_1 needed to achieve critical inversion is very small [31].

Transitions to A are generally assumed to be from the lowest level of T_1 . These transitions may be either radiative, called phosphorescence, or, as is the situation for most organic dyes, nonradiative. The longer T_t is, the faster the dye pumping pulse has to be to excite

enough molecules to reach critical inversion before the T₁ trap prevents dye laser emission.

The T_1 trap problem has caused a lot of research to be done to find substances which will shorten T_t via collisions with molecules in T_1 . This action is called quenching and two very effective quenching agents are cyclooctatetraene (COT) and molecular oxygen.

B. DYE LASER MATHEMATICS

1. Dye Laser Gain Equation

The dye laser gain equation is a mathematical statement of the production and loss rates for photons in the laser cavity. Photons are produced by stimulated emission from the excited singlet state to the ground singlet state. Photons are lost by $S_0 - S_1$ and $T_1 - T_2$ absorption and by loss through extrinsic means such as output mirrors, optical surface scattering, etc. Photon production is also possible by stimulated emission from the excited triplet state but nonradiative decay is so rapid from this state to the lower triplet state that this source of photons can be neglected.

A simplified schematic diagram of a dye laser is shown in Figure 2. L_1 is the dye cell length, R_1 and R_2 are the input and output mirror reflectivities, T_1 and T_2 are the dye cell window transmittances, and n_0 , n_1 , n_2 , etc. are the photon densities at any time at the indicated positions of a round trip in the laser cavity. This particular cavity is used as the model for the following derivation of the gain equation.

The rate of photon production per unit length in the dye may be written as follows:

$$\frac{dn}{dl_{total}} = \frac{dn}{dl_{emission}} - \frac{dn}{dl_{absorption}}$$
(1)

and the second second





The first term on the right hand side of (1) is photon production by stimulated emission and the second term is photon loss by singlet and triplet state absorption.

Yariv and Gordon [32] have shown that the expression for the transition rate caused by a monochromatic beam of light of wavelength λ is

$$\frac{\mathrm{dn}}{\mathrm{dt}_{\mathrm{e}}} = \frac{N_{\mathrm{s}}\lambda^{4}\mathrm{E}(\lambda)\mathrm{n}(\lambda)}{8\,\mathrm{\pi}\,\mathrm{T}\,\mathrm{s}^{-3}}\,\mathrm{cm}^{-3}\,\mathrm{sec}^{-1} \tag{2}$$

where N is the population density of the excited singlet state, $n(\lambda)$ is the number of photons of wavelength λ per cm³ of active medium, T is the excited singlet state lifetime, \overline{n} is the index of refraction, and $E(\lambda)$ is the fluorescence lineshape function normalized so that

$$\int_{O}^{\infty} E(\lambda) d\lambda = \emptyset = \text{quantum yield} = \frac{\text{photons emitted}}{\text{photons absorbed}}$$

An expression for the first term of (1) can now be derived. Since

$$\frac{dn}{dt}_{e} = \frac{dn}{dl}_{e} \times \frac{dl}{dt}$$
 and $\frac{dl}{dt} = \text{velocity} = \frac{c}{n}$

where c is the speed of light, it follows that

$$\frac{\mathrm{dn}}{\mathrm{dl}_{\mathrm{e}}} = \frac{N_{\mathrm{s}} \lambda^{4} \mathrm{E}(\lambda) \mathrm{n}(\lambda)}{8 \, \mathrm{m} \, \mathrm{c} \, \mathrm{T}_{\mathrm{s}} \, \mathrm{\bar{n}}^{2}} \quad . \tag{3}$$

The molecular extinction coefficient, which is the absorption eross-section for a single dye molecule, can be used to write the singlet and triplet absorption loss per unit length for (1). These terms are

$$\frac{dn}{dl}_{singlet} = n(\lambda) N_{o} \sigma_{s}(\lambda)$$
(4)

$$\frac{dn}{dl}_{\text{triplet}} = n(\lambda) N_{t} \sigma_{t}(\lambda)$$
(5)

where N_o and N_t are the population densities of the ground singlet and triplet states, and σ_s and σ_t are the wavelength dependent singlet and triplet absorption cross-sections.

Let

$$\sigma_{e}(\lambda) = \frac{\lambda^{4} E(\lambda)}{8 \pi c T_{s} \overline{n}^{2}}$$

and substitute (3), (4) and (5) into (1). This yields the total production rate of photons per unit length in the dye medium which is

$$\frac{dn}{dl}_{total} = n(\lambda) \sigma_{e}(\lambda)N_{s} - n(\lambda) \sigma_{s}(\lambda)N_{o} - n(\lambda) \sigma_{t}(\lambda)N_{t}.$$
 (6)

Separation of variables of (6) and integration over the length of the dye cell with the boundary conditions that where 1 = 0, $n = n_1$, and where $1 = L_1$, $n = n_2$, yields the solution

$$n_2 = n_1 \exp \left[\sigma_e N_s - \sigma_s N_o - \sigma_t N_t \right] L_1.$$
(7)

The n's and σ 's are still wavelength dependent and from Figure 2 it can be seen that since the paths traveled by photons n_1 and n_5 are through the same medium but in opposite directions that (7) also holds for the relation between n_6 and n_5 .

It is obvious from Figure 2 that $n_1 = n_0 T_1$, $n_3 = n_2 T_2$, $n_4 = n_3 R_2$, $n_5 = n_4 T_2$, $n_7 = n_6 T_1$, and $n_8 = n_7 R_1$. Substitution of these relations consecutively into (7) gives the result

$$n_8 = n_0 \exp[\sigma_e N_s - \sigma_s N_o - \sigma_t N_t + \frac{1}{2L_1} \ln(R_1 R_2 T_1^2 T_2^2)] 2L_1.$$
(8)

Let $G(\lambda)$ be the equation in brackets in (8) and define it to be the gain. Then

$$n_8 = n_0 \exp[2 G(\lambda) L_1]$$
, and

$$G(\lambda) = \sigma_{e} N_{s} - \sigma_{s} N_{o} - \sigma_{t} N_{t} + \frac{1}{2L_{1}} \ln \left[R_{1} R_{2} T_{1}^{2} T_{2}^{2} \right]$$
(9)

which is the desired gain equation and is similar to the results obtained by Snavely [8] and McColgin, et al. [33].

2. Dye Laser Rate Equations

The dye laser will lase at the value of λ where (9) is a maximum. To find this wavelength it is necessary to solve the state population density rate equations for the dye laser. These equations are

$$\frac{\mathrm{dN}}{\mathrm{dt}}\mathrm{s} = -\frac{1}{\mathrm{T}}\mathrm{s}^{\mathrm{N}}\mathrm{s} + \mathrm{P}(\mathrm{t})\mathrm{N}_{\mathrm{o}}$$
(10A)

$$\frac{dN}{dt}t = -\frac{1}{T_t}N_t + k_{st}N_s$$
(10B)

$$\frac{dN}{dt}o = -P(t)N_{o} + (\frac{1}{T}_{s} - k_{st})N_{s} + \frac{1}{T}_{t}N_{t}$$
(10C)

$$N = N_{o} + N_{s} + N_{t}$$
(10D)

N is the total population density and (10D) is true only for an enclosed system undergoing no photochemical processes. N_t , N_o , and N_s are as defined for the gain equation, T_s and T_t are the lifetimes of the singlet and triplet states, k_{st} is the excited singlet to ground triplet intersystem crossing rate constant, and P(t) is the optical pumping rate.

There are several assumptions made in writing the rate equations in the form of (10A), (10B), (10C) and (10D). These assumptions are:

(a) The number of molecules involved in intersystem crossing from the triplet to the singlet state is negligible.

(b) The molecules initially excited into the first excited singlet state reach thermal equilibrium in a short time compared to the pumping time.

(c) Ground triplet to excited triplet state absorption is not negligible, but the population of the excited triplet state is negligible because of rapid nonradiative decay.

(d) The dye laser has essentially only three levels. (The fine vibrational and rotational levels in each main state are neglected.)

(e) The effect on state population densities caused by dye self-absorption of the lasing wavelength is negligible. (There are approximately two orders of magnitude difference between the absorption coefficient for the pumping wavelength and the expected lasing wavelength so this assumption is not too restrictive.)

(f) The point of maximum gain is reached so fast that the effect of reduction of the excited singlet state due to stimulated emission can be neglected. (This assumption has been shown to be realistic by Atkinson and Pace [34] and is very true at the onset of lasing; however, as lasing progresses and photon buildup occurs in the laser cavity the stimulated emission term can become very significant. The same is true of the self-absorption term since it also depends on intracavity power.)

Equations (9), (10A), (10B), and (10D) are used to evaluate the dye laser model in a computer simulation in Section IV. Equation (9), the gain equation, is used in the discussion of theoretical results to explain the theoretical time dependent wavelength sweep of the dye laser.
III. EXPERIMENTAL PROCEDURES AND RESULTS

A. EXPERIMENTAL SET UP

The apparatus used in generating and measuring the output of the Argon and dye lasers was set up as shown in Figure 3. A Beck Wavelength Reversion Spectroscope with a single line resolution of approximately four Angstroms was also used to monitor the dye laser output wavelength.

The beam splitter indicated in Figure 3 was actually an integral part of the front end of the Argon laser and is normally used as part of a power monitoring system. This system was modified so that the Argon laser output pulse could be monitored on an oscilloscope via a photodiode detector.

The diode detector and power meter were both tested for linearity and saturation at higher powers than could be expected at their locations in Figure 3. They both exhibited no signs of approaching saturation and their deviations from linearity over the testing range were minimal.

The risetime of the Argon laser is normally limited to seven nanoseconds because of cable transit times, etc., however, in the particular set up of Figure 3 the risetime was limited to 12 to 15 nanoseconds because that was the lower limit of the pulse generator risetime. An explanation of the cavity dump mechanism can be found in Maydan [35].

B. EXPERIMENTAL PROCEDURES

The first step in the procedure was the alignment of the Argon laser cavity dump system. This system consisted of a curved front half-mirror with vertical and horizontal adjustments, a rear curved mirror with vcrtical and horizontal plane tilt controls, and an acoustooptic cell housing



!

with controls to adjust its longitudinal and vertical positions as well as its angular alignment. All of these controls had to be precisely set for the cavity dump system to function properly, and the Argon laser power output and pulse shape had to be constantly monitored because some of the adjustments were extremely delicate. A slight movement of the rear mirror adjustments, for example, severely degraded the Argon laser output pulse and beam mode structure.

The next step in the procedure was the alignment of the Argon and dye lasers. This was done on a specially designed platform which had permanent hold down clamps for the Argon laser and kinematic mounts for the dye laser. Alignment of the two lasers was very sensitive and was a critical part of the experimentation since any nonparallelism of the dye and Argon laser cavities severely reduced the power conversion efficiency.

The Argon laser power supply was set for a current of 30 amperes after alignment was achieved. With the pulse generator delivering three volt, thirty nanosecond pulses at a repetition rate of one megahertz to the cavity-dump rf power supply, the average power output of the Argon laser was 500 milliwatts. These values and settings were then used as Argon laser operating reference points and standards for the rest of the experimentation.

The final step in the procedure was to monitor the output of the Argon and dye lasers. The Argon and dye laser powers were recorded, the Argon laser pulse shape was sketched, and the various dye laser wavelengths noted as the pulse width of the pulse generator was varied.

the second se

C. EXPERIMENTAL RESULTS

The results yielded some new and interesting properties of the dye laser. The wavelength of peak output power and efficiency of the cavitydumped laser-pumped dye laser was found to occur at a shorter wavelength than that of the continuous wave laser-pumped dye laser. In addition, the tuning range of the pulsed dye laser was narrowed.

The normalized output for both the continuous-wave laser pumped and pulsed-laser pumped dye laser are shown for comparison in Figure 4. The same dye laser was used to obtain both curves.

A two piece mathematical model which is an excellent fit to the actual Argon laser output pulse used to obtain the pulsed dye laser curve is shown in Figure 5. The repetition rate of this pulse was one megahertz.

It was also discovered that as the pulse width driving the cavitydumped rf power supply was lengthened the wavelength of peak emission shifted toward longer wavelength and the laser efficiency decreased. In addition, for a given pulse width, an increase in the output power of the Argon laser (accomplished by increasing the current to the Argon laser power supply) shifted the peak emission wavelength to shorter wavelengths.

The lowest peak wavelength observed was 5650 Angstroms and the highest conversion efficiency obtained was almost thirty percent. This same dye laser pumped with a continuous wave laser had an efficiency of only seventeen percent at the peak output wavelength of 5800 Angstroms.

The fluorescence spectrum of the dye in the laser was also measured by use of a spectroscope with a high intensity white light source. This spectrum along with the wavelength-power curve of the cavity-dumped laserpumped dye laser are compared in Figure 6. It is obvious from this figure that the peak lasing wavelength for the pulsed pump is approaching the peak of the dye fluorescence curve.









Figure 5: Typical Argon laser output pulse.

.









A summary of the significant experimental observations is as follows:

(1) The pulsed dye laser wavelength spectrum exhibits a substantial green shift from the spectrum of the CW pumped dye laser.

(2) The shift increases and approaches the dye fluorescence curve as the pulse length from the cavity-dumper gets shorter.

(3) The efficiency is greater and the tuning range shorter for the pulsed dye laser.

IV. DYE LASER MODEL EVALUATION

Experimental results showed that the output wavelength spectrum of the cavity-dumped laser-pumped dye laser approached the dye fluorescence curve. The dye laser model of Figure 2 is analyzed in this section using equations (9), (10A), (10B) and (10D), which are the model gain and rate equations, and the pumping waveform of Figure 5 to determine if the spectrum shift can be predicted.

A. LITERATURE SURVEY

The first application of applying rate and gain equations to the dye laser was made by Sorokin, et al. [36, 22]. Their application assumed a Gaussian waveform pumping pulse and they attempted to predict the transient pulse shape of a pulsed Chloro-aluminum phthalocyanine (CAP) dye laser. They concluded from their results that a rapid pulse risetime was necessary or triplet state accumulation could prevent dye laser action.

The second analysis of Sorokin and his group again assumed a Gaussian pumping pulse and attempted to predict the efficiency of various dye lasers as functions of quantum efficiency and the excited singlet to triplet intersystem crossing rate constant (k_{st}) . The results from this work yielded the prediction that dye laser efficiency should increase as the intersystem crossing rate time constant decreased or as the quantum efficiency increased.

A year later, in 1968, Bass, Deutsch and Weber [37] used the rate and gain equations to predict lasing frequencies and time to lasing for Gaussian shaped laser and flashlamp pumped dye lasers. They had

experimentally observed that a laser pumped dye laser emitted a shorter wavelength than when flashlamp-pumped, however, different laser cavities were used for the two pumping conditions and this led to different parameters entered in the gain equation and to the conclusion that differing cavity Q's accounted for the higher lasing wavelength of the flashlamp pumped case.

The following year Bass and Weber [38] again used dye laser rate and gain equations. This time they assumed a rather long Gaussian shaped pumping pulse and included triplet state effects in their discussions. They still concluded that cavity Q was the main determinant of lasing wavelength, however, they conceded that determination of lasing wavelength could be more complex if triplet state effects were significant.

In mid 1970 Keller [39] solved for pseudo steady state solutions to the rate equations (all time derivatives were set equal to zero) to investigate the effects of quenching agents on singlet and triplet state populations. He arrived at the conclusion that a specific quencher for the triplet state would markedly improve laser efficiency and that a substance that quenched both singlet and triplet states would usually improve the efficiency.

The next group to use rate equations was McColgin, et al. [33]. They considered triplet state effects but only under steady state conditions as is found in some continuous wave dye lasers, and they concluded that the emission wavelength adjusts itself so that the ratio of mirror losses, scattering losses, etc. to singlet absorption losses remains approximately constant.

Pappalardo, Samuelson and Lempicki [40] in 1972 used the rate equations to calculate the efficiency of dye lasers as a function of pump

parameters and triplet state lifetime for pump pulses in excess of one microsecond in duration. They assumed a Gaussian shaped pumping pulse and predicted that long pulse operation of up to 30 microseconds was feasible for short triplet state lifetimes but that the efficiency would go down as the pumping pulse was lengthened.

The most recent use of rate equations has been by Strome and Tuccio [17] who used results to improve their original dye laser [16], by Atkinson and Pace [34] who used them to calculate the lineshape of a Fabry-Perot etalon tuned Rhodamine 6-G dye laser, and by Streifer and Saltz [41] who used the equations to analyze an acoustooptically tuned dye laser.

B. GAIN AND RATE EQUATION PARAMETERS

The evaluation of the dye laser system rate and gain equations for a cavity such as the one in Figure 2 requires a lot of physical data. The actual dye used to obtain the experimental data had a mixture of methanol and water as the dye solvent; however, there is no information available on the dye in this particular solvent hence physical data available for ethanol solutions was used. It was assumed that any predictions or conclusions based on the results would apply to Rhodamine 6-G in the actual solvent. (The main difference in the solvents is that both the singlet absorption and fluorescence spectrum peaks are further in the green region of the visible spectrum when the solvent is 100 percent ethanol.)

The fluorescence lineshape and singlet absorption cross section of a 10^{-4} Molar solution of Rhodamine 6-G in ethanol are shown in Figure 7. These curves were obtained by F. Grum of Kodak Research Laboratories and were reproduced in Reference 8. Snavely [8] measured the fluorescence yield of the dye and obtained a value of $\emptyset = 0.83$ and the fluorescence spectrum of Figure 7 is normalized to this value; i.e.

$$\int_{-\infty}^{\infty} E(\lambda) d\lambda = \emptyset = 0.83.$$

There is no data available on the triplet absorption cross section for an alcohol solution of Rhodamine 6-G. The only available curve is shown in Figure 8 which was determined by Buettner [42] using a flash photolysis technique for Rhodamine 6-G in polymethyl methacrylate.

Several values of singlet and triplet state lifetimes are given in the literature. The various values given for T_s are 7.4 nsec in Weber and Bass [38], 4.8 nsec in McColgin, et al. [33], 7.3 nsec in Snavely and Peterson [31], and 5.5 nsec in Mack [43]. The reason for the use of different values by the groups mentioned above is due to different measured values. An average value of 6.3 nsec was used in this analysis because there were no significant reasons why one value should be preferred.

There are also many values of T_t given in the literature, but the best evidence available suggests that the most reasonable value is 100 nsec. Keller [39] arrived at this value in his analysis of Oxygen quenching in an alcohol solution and Snavely and Schafer [9] obtained 100 nsec as a measured value.

The value of the intersystem crossing rate constant, k_{st}, can be determined from the formula

$$k_{\rm st}^{-1} = T_{\rm s} \phi / (1 - \phi)$$
 (11)

which is found in Snavely [8] and others. The value of 0.84 from Bass and Steinfeld [44] is the most common value found for \emptyset and since the value of 6.3 nsec is assumed for T_s, (11) yields the value of 29.4 nsec for k_{st}⁻¹. The fluorescence spectrum of Figure 7 is normalized using \emptyset = 0.83 so that spectrum and equation (11) are compatible.







The easiest constant to evaluate is N, the total molecular density. The actual dye solution was 3 x 10^{-4} Molar so that

$$N = 3x10^{-4} \text{ Molar x } \frac{1 \text{ gram-mole}}{1 \text{ Molar-liter }} \times \frac{6.023x10^{23} \text{ molecules}}{\text{gram-mole}} \times \frac{1 \text{ liter}}{10^3 \text{ cm}^3}$$

which is 1.8×10^{17} molecules/cm³. This assumes that the density of the solvent at room temperature is approximately the same as at the standard reference temperature.

The values of $L_1 = .1$ cm, $R_1 = .995$ and the curve of R_2 versus wavelength were obtained from the manufacturer [45]. The dye cell windows were anti-reflection (AR) coated so typical values for AR surfaces of .98 were used for T_1 and T_2 .

The remaining constants needed for evaluation of $\sigma_e(\lambda)$ are c, the speed of light, and \overline{n} , the index of refraction. The value used for the speed of light was 3×10^{10} centimeters/sec and the index of refraction of the dye solution was approximately 1.33.

The optical pumping rate, P(t), may be written as $\sigma_p I_p/hf_p$ where σ_p is the dye absorption cross section for f_p , the pump frequency, h is Plancks constant and I_p is the pump irradiance in watts/cm². The pump wavelength was 5145 Angstroms and the corresponding frequency was 5.82 x 10^{14} sec^{-1} . Plancks constant has a value of $6.625 \times 10^{-34} \text{ sec}^{-1}$. The value of $2.37 \times 10^{-16} \text{ cm}^2$ obtained from Figure 7 was used for σ_p .

Pump irradiance can be written as power per unit area. The model of the pump power as a function of time is shown in Figure 5. The average power output of the Argon laser was 0.5 watts and the value of 42.7 watts for P_0 was obtained from the formula

$$P_{average} = \frac{1}{T} \int_{o}^{T} P(t) dt.$$

There is a problem in defining the effective area and volume in which the pump beam is absorbed. Jacobs, Samuelson, and Lempicki [46] assumed uniform pumping in their analysis of losses in a continuous wave dye laser and used the beam waist area as the effective pumped area. At least some of the Argon laser beam must exit the dye volume before uniform pumping can be assumed [40], otherwise the amount of beam penetration into the dye and the size of the pumped volume cannot be determined. The dye laser used in the experiment allowed approximately five percent of pump power to exit the dye cell [45], therefore, uniform pumping was assumed. The beam waist radius was six microns [45] so the assumed effective area was $1.13x10^{-6}$ cm².

The assumption of uniform pumping is one of the most marginal in dye laser analysis. The complexities introduced by trying to evaluate modematching of the pump and dye laser beams in the dye cell and nonuniform excitation of dye molecules, however, makes the assumption practical.

C. GAIN AND RATE EQUATION SOLUTIONS

1. Solution Method

The first step in the theoretical analysis was the solution of the rate equations for the state population densities as a function of time. The solution was carried out using a program entitled INTECl of the U.S. Naval Postgraduate School Computer Library. (See Appendix A) INTECl uses a fourth order Runga-Kutta method with programmable step size changes to achieve the solution to simultaneous ordinary differential equations. The only modifications made to the INTECl program were those necessary to enable it to handle the wide range of parameter values used in the rate equations.

The values of the state population densities were then used to evaluate the gain equation. This was accomplished by first selecting values of state densities at a particular time and then evaluating the gain using the wavelength dependent parameters over a large range of wavelengths. The succeeding steps followed the same pattern and this was continued until the point of maximum gain versus time and wavelength reached a constant relationship; i.e., the wavelength of maximum gain remained unchanged with time.

The purpose of evaluating the gain in this manner was that the lasing wavelength of the laser model should be where the gain is maximum. This was then compared to the dye fluorescence spectrum to see where the lasing wavelength was in relation to the peak wavelength of the spectrum.

The last step was to predict where the experimental laser should lase based on the theoretical comparison and the fluorescence spectrum of the experimental dye. Theoretical and experimental results should agree if theory was sound.

2. Theoretical Results

A graph of the peak gain wavelength versus time is shown in Figure 9. It can be seen from the figure that lasing starts at the longer wavelengths and sweeps toward the shorter wavelengths. Figure 9 also contains a graph of the ratio of triplet state population to ground singlet state population (N_t/N_o) versus time.

The results of evaluation of the gain versus wavelength at steady state is shown in Figure 10. It can be seen that the peak wavelength (5570 Angstroms) corresponds to the peak wavelength of the Rhodamine 6-G fluorescence curve of Figure 7.

The results displayed in Figure 9 and the gain equation, equation (9) of Section II, can be used to explain the theoretically obtained dyc



Peak gain wavelength and ratio of triplet to ground singlet state population for 3x10⁻⁴ Molar Rhodamine 6-G laser model. Figure 9:




Figure 10: Theoretical normalized gain versus wavelength for Rhodamine 6-G dye laser model.



laser wavelength sweep. The two wavelength dependent losses in the gain equation are caused by singlet and triplet absorption. The ratio of N_t/N_o at the onset of lasing is very small indicating domination of the singlet state absorption loss. (Singlet absorption is associated with N_o and triplet absorption with N_t .) Figure 7 shows that singlet absorption is greatest for shorter wavelengths, hence there should be a tendency for long lasing wavelengths at the onset of lasing.

As pumping continues the ratio of N_t/N_o becomes greater. This indicates that as time passes there should be a tendency for triplet absorption losses to dominate. Figure 8 shows that triplet absorption is greater at the longer wavelengths (up to 6000 Angstroms which is within the lasing range of Rhodamine 6-G) and, therefore, the dye laser should end up lasing at wavelengths closer to the peak of the dye fluorescence curve. (The above statements are true in this case for the specific pulse model of Figure 5. Pulses of different shape, duration, and intensity may or may not cause the sweep.)

V. CONCLUSIONS

A. EXPERIMENTAL AND THEORETICAL COMPARISON

The theoretical analysis predicted that for the pumping pulse model of Figure 5 and after a wavelength sweep of very short duration the dye laser would lase at or near the peak of the dye fluorescence spectrum. It was observed experimentally that the dye laser lased approximately 50 Angstroms from the peak of the dye fluorescence curve in the pulsed case (Figure 6) rather than 200 Angstroms as in the continuous wave pumped case (Figures 4 and 6). It may be concluded, therefore, that theory and experimentation are in agreement within the limits of the assumptions made in the study.

The variable of least confidence in the entire analysis was the experimental dye fluorescence curve. The peak of the fluorescence curve for Rhodamine 6-G in an ethanol solution is typically about 5570 Angstroms [8] and in a water solution is about 5750 Angstroms [46]. The actual solvent was determined by nuclear magnetic resonance measurements to be approximately two-thirds water and one-third mathanol. This should have caused the fluorescence peak of the dye to be somewhere around 5660 to 5700 Angstroms which is much closer to the observed peak wavelength than the 5600 Angstrom peak of the fluorescence curve of Figure 6. One explanation for the difference between the measured and expected fluorescence peaks is the fact that the spectrometer calibration curve used to measure the experimental dye fluorescence spectrum had a steep slope and was very sensitive to spectrometer dial readings.

B. REMARKS

The decrease in efficiency of the dye laser as the pulse length was widened can be explained very simply in terms of the rate equations. The lengthening of the pulse decreased the peak power of the pumping pulse which in turn caused a longer time to reach the onset of lasing. As a result, the triplet state had more time to build up and extract molecules from the singlet lasing system - efficiency had to decrease. There are several things which hinder the analysis of the dye laser. There is not much data available on dye properties in various solvents and the properties of the triplet state are still pretty much unknown. There is also a lot to be learned about the effects of thermal heating and nonuniform pumping of the dye and no one as yet has begun to investigate the effect of chemical reaction and relaxation rates on the dye laser. Stability and noise considerations also require investigation.

APPENDIX A

DYE LASER RATE EQUATION COMPUTER PROGRAM LISTING

The computer listing on the following pages contains the program for the solution of the dye laser rate equations. The main part of the program is SUBROUTINE INTEG1 and the comments preceding its operational statements completely describe its use. The rate equations in the program correspond to equations (10A), (10B) and (10D) of Section II. The data cards used in the actual solution and described in the usage comments of INTEG1 are located at the end of the listing.

INTEGL had to be modified to some extent so it could handle a wider range of parameter values than was originally allowed. These modifications included input and output format and an increase in the number of allowed integration steps and pages of output, but no changes were made to the basic fourth order Runga-Kutta integration and other computational algorithms.

- - -

DYE LASER RATE EGLATION SCLUTION

THE OPULATION u, AND O FUNCTION ഗ II. EQUATIONS II. IN THE N; X(2) IS T ET STATE POP ł ⊲• 42*Y*X(1) Ч Ч С Ч С С PCWEP A THE THREE EQUATIONS BELOW ARE THE DYE LASER RATE CRRESPOND TO EQUATIONS 10A, LOB AND 100 CF SECTION II GUATIONS X(1) IS THE EXCITED SINGLET STATE PCPULATION RIPLET STATE POPULATION AND X(3) IS THE GROUND SINGLET ŝ. \sim 1 ЦО ŧ E ARGON LASEI GMENT CURVE VII. -3 . ග • 1.0 Ľ FS SNYDE // SNY52CE3 JOB (2083,029C,DC12). SNYDE // FCRT.SYSIN DD * EGION.GC=100K DIMENSICN X(30),XDGT(30),C(15) C(10)=1,0 C(10)=1,0 C(11) = 100 C(11) = 100 C(12) = 100 C(12) = 100 C(11) = 100 C(12) = 100 C(11) = 100 C(11) = 100 C(12) = 100 C(11) = 100 C(12) = 100 C(11) = 1 (Z) BELON IS WITH THE TWO -UNCTION PGWER(2) FF(2.6E.12.0) GO TC 10 CKER = (42.7*Z)/12.0 CKER = 42.7*EXP((12.0-Z)/6.0) ETURN NC αш E IN ACCORDANC VI I пысакш OUF21 000000000 0000



1 N T 1 0 0 1 0	INT10050	INT10060 INT10060 INT10060 INT100080 INT100080 INT100100 INT10120	INTI0150 INTI0150 INTI0160 INTI0180	INTI0200	INT102300 INT102300 INT10240	INT10260 INT10260			INT10330				INT10420	INT10450	INT10460
<pre>SLEROUTINE INTEG4(/TC/,/XC/,/DX/,/C/)</pre>	SLBROUTINE INTEGI (MODIFIED VERSICN CALLEC INTEG4) PLRPOSE	INTEGI PROVIDES A SIMPLE BUT FLEXIBLE MEANS FOR RUNGA-KUTTA SOLUTION OF ORDINARY DIFFERENTIAL EQUATIONS. INPUT AND OUTPUT ROUTINES ARE BUILT-IN ALONG WITH PRINTER PCINT FLOTTING RCU- TINES. THE INTEGRATION STEP-SIZE FOR THE FOURTH CRDER NUMERI- CAL INTEGRATION CAN BE CHANGEC BY APPRCFRIATE CATA INPUT AND/CR PROGRAMMING. THE SUBRCUTINE IS USUALLY USED FOR EQUA- TIONS WITH PREDETERMINED PARAMETERS ELT CAN BE USED TO SCUVE	NUKE CUMPLEA PRUDLEMS. USAGE INTEGI HAS BEEN COMPILEE ANG ADDED TO THE FORTRAN LIERARY Andrien to the ter one heer chivy shentt a standar (27220)	PROGRAM DECK AS FOLLOWS:	<pre>// STANDARD JOB CARD // EXEC FORTCLG,REGION.GO=100K //FCRT.SYSIN DD # DIMENSICN X(30),XDJT(3C),C(15)</pre>	C(10)=1. 1 CALL INTEG1(T,X,XCJT,C)	* * * * * * * * * * * * *	* FORTRAN EQUATIONS TO BE SOLVEC *	· · · · · · · · · · · · · · · · · · ·	GO TO 1 END	//GC.SYSIN DD *	* * * * * * * * * * * * * * * * * * *	* DATA CARDS AS DESCRIEEC BELCW	有 开 开 开 开 子 子 子 子 子 子	/* (DRANGE)

•

41



NT10								
			-d hand hand hand hand hand have	a territoria de la constructiona de la construction de la construction de la construction de la construction de La construction de la construction d		and hand bend helds have be		
ECUATION STATEMENTS	THE NUMBER OF EQUATIONS TO BE SOLVED, "N", MUST BE LESS THAN OR EQUAL 30. THEY MUST BE RECUCED TO STANDARD FIRST CREER FORM AND THEN EXPRESSED IN FERTRAN. ACCITIONAL NENSUBSCRIPTED VARIABLES MAY BE USED AS WELL AS STANCARD LIBRARY FUNCTIONS VANDAR SUBROUTINES. THE ORDER OF THE ECLATIONS IS USUALLY UNIMPORTANT. A TYPICAL SET MIGHT REAC:	<pre>ERROR=SIN(C(1)*T)-X(1) XDOT(1)=X(2)1*X(1) XDOT(2)=ERROR+C(2)*XDOT(1)</pre>	THE X ARRAY CONTAINS THE DEPENDENT VARIABLES. THE XCGT ARRAY CONTAINS THE FIRST DERIVATIVE WITH RESPECT TO THE INCEPENCENT VARIABLE ("T" IN THIS EXAMPLE). THE ECLATIONS TO BE SUPPLIED BY THE USER DEFINE EACH "XDCT" IN TERM'S OF THE INDEPENCENT AND DEPENDENT VARIABLES, CONSTANTS AND CTHER VALUES IN THE "XDOT" ARRAY.	CONSTANTS (C(I), ILE.8) ARE READ FROM A CATA CARD. CCNSTANT NUST NEVER BE USED EXCEPT AS IN THE SAMPLE DECK SETUP ABOVE. CONSTANTS C(I1), C(L2), AND C(I2) CCNTRGL FREQUENCY CF PRINT-CUT DURING INTEGRATION, THE FREGLENCY OF PLOT POINT GENERATION, AND THE INTEGRATION STEP-SIZE (IN THAT GROER). THE DEFAULTS ARE PRINT-GUT EVERY 20TH STEP-SIZE (IN THAT GROER). GENERATED AT EACH 5TH STEP. THE INTEGRATICN STEP-SIZE IS NENTS ABOVE (AT ANY POINT DURING THE SCLUTICN) SUCH AS:	C(11)=10. C(12)=1.	CAUSES PRINT-CUT TO TAKE PLACE AT EVERY 10 STEPS AND PLOT POINTS TO BE GENERATED AT EACH STEP. NOTE THAT NO MORE THAN 4500 INTEGRATION STEPS, 450 LINES OF PRINT-OUT, CR 900 PLCT POINTS CAN BE GENERATED IN ANY ONE RUN.	ANY LEGAL FORTRAN STATEMENT CR TECHNIGUE CAN BE USEC IN THE CALLING PROGRAM, PRCVIDED THAT A TRANSFER OF CCNTROL DCES NOT PREVENT THE PROCESSING OF ALL RELEVANT STATEMENTS CURING EACH STEP OF THE INTEGRATION. FOR EXAMPLE THE FOLLOWING WCLLC BE ALLOWED:	IF(T-10.0) 3,2,2 3 C(1)=0. 2 CONTINUE



nd brind											
WHICH CAUSES C(1) TO TAKE THE VALUE SPECIFIED CN THE CATA CARDI FOR T.LT.10. AND THE VALUE 0. FCR T.GE.10. ONLY CUANTITIES X(1) AND T CAN BE CUTPUT. IF SCME QUANTITY CTHER THAN THE INDEPENDENT VARIABLE, DEPENDENT VARIABLE CR DERIVATIVE IS CESTRED AS CUTPUT, IT MUST EE EQUATEC TC AN X(1) SUCH THAT IN N.LT.L.LE.30. SUCH AS:	X(3) = ERRCR X(4) = ERRCR*ERKOR.	ALL SUCH AUXILLIAFY EQUATIONS MUST BE PLACED BETWEEN THE "CALL INTEGI" AND "GO TO 1" STATEMENTS.	ATA CARDS	IRST: CONTAINS A JOB IDENTIFICATION LAGEL PUNCHEC IN CCLS I TC 48. IT WILL BE REPRODUCEC IN THE FRINT-CUT.	ECOND: CONTAINS THE NUMBER OF RUNS TO EE PROCESSEC (.LE.S) PUNCHED IN COL 1. THE RUN NUMBER IS PLACEC ON ALL CUTPUTI ALONG WITH THE ASOVE JCB IDENTIFICATION.	HIRD: CONTAINS THE VALUE OF N, THE NUMBER OF ECLATICNS TO BE I SOLVED. PUNCHED RIGHT-JUSTIFIED IN COLS 1 AND 2. WHEN N IS LESS THAN 10, IT MUST BE PUNCHED IN COL 2. WHEN IT IS EQUAL TO OR GREATER THAN 10 IT MUST BE PUNCH-I ED IN COLS 1 AND 2. IT MUST BE LESS THAN OR EQUAL TO 30.	CLRTH: INTRODUCES THE INITIAL AND FINAL TIMES CF T (TI & TF) I TOGETHER WITH THE STEP-SIZE(S), CT. THE INTEGRATION CAN I BE PROCESSED IN UP TO 3 SEGMENTS, EACH WITH A CIFFERENT I STEP-SIZE, THUS THE DATA CARD CAN CONTAIN:	TI - CTI - TF TI - DTI - TFI - CT2 - TF OR TI - CTI - TFI - CT2 - TF2 - CT3 - TF	THE CORRESPONDING DATA VALUES ARE PUNCHED (WITH DECI- Mals), IN THE ABOVE ORDER, IN CCLS 1-10,11-20,21-3C, ,61-70.	IFTH: CONTAINS THE CONSTANTS C(L) THRL C(8) PUNCHED (WITH DECIMALS) IN COLS 1-10,11-20;,71-80, ELANK MAY BE USED FOR EITHER ZERD CR UNUSED CCNSTANTS.	IXTH: CONTAINS THE INITIAL CCNCITIONS (X(1)(0)) THRCUGH I (X(N)(0)) IN THE SAME FCRMAT AS THE FIFTH CARC. ACCI- I
			()	u.	0)	F-	L.,			LL.	0

-

.



. .



INTI15400 INTI15400 INTI15500 INTI15500 INTI15500			1 1 1 2 2 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0	IN1122400	INTI23200 INTI23200 INTI23200 INTI23200	INT 12350 INT 12350
PRCGRAMMER J.R.WARD, GCTCBER 1963 REVISED 6/64,6/65,6/67,11/71 REAL*8.ITITLE(12),JTITLE(3),IBLANK/	<pre>1KTITLE(6) DIMENSIGN x(20), Bx(3C), XC(3C), C(15), IP(40), IG(10), FR(10), GR(1C), 1Tx(5), TY(5), X1(900), Y1(900), X2(5CC), Y2(9CC), X3(900), Y3(5C0), 2X4(500) Y4(50C) 2X4(500) Y4(50C) 2X4(500) Y4(50C) 2X4(500) Y4(50C) 1x(5), TY(5), TUN*, ', ', ET(9)/' 1 *, 2 *, 3 *, 4 1x(5), TY(5), TUN*, ', ', ', ', ', ', ', ', ', ', ', ', ',</pre>	I READ(5,100) (ITITLE(1), I=1,6) C READ (5,101)NR C1 READ (5,101)NR C1 READ (11) C2 REMAT (11) C2 REMAT (12) ARC = 0 C0 TC 1000 C0 TC 1000	<pre>ARITE(57200) UST NCT EXCEED 30. D FCRMAT (7/7/,43H ERROR IN ORCER CF EQUATION. MUST NCT EXCEED 30. STCP C NRC = NRC + 1 . WRITE (6,201),(JTITLE(1),1=1.6)</pre>	I FLEMAT VITII/////////////////////////////////	<pre>5 htile(6/203) 5 format (//37x,21HCNE RUN IS CALLEC FOR ,///,18H INPUT DATA RECCRD 0C TO 7 6 hrite(6,204)NRC 04 fcrmat (///,34H INPUT DATA RECORE FOR RUN NUMBER ,11)</pre>	7 WRITE(0,205)NN C5 FCPMAT (//,22H ORDER OF EQUATIONS = ,12,/) Read (5,103)T1,DT,TF1,CT2,TF2 33 FCFMAT(5F15,6) 33 Read (5,5000) DT3.TF3
•		101	10.0	502	202	10

.

1 N T 1 23 CU	INT12410	INT12440	INT12460	INT124500	INT12530 INT12550 INT12550		00000000000000000000000000000000000000	INT12630	111126600 111126600	INT12690 INT127690 INT127690	INT1274	INT12760	INT127500	INT12820
-CRMAT(2F15.6)	FF(DT2'NE.0.) GO TC 9 RITE(2.206)T17F FCRMAT (22H INITIAL TIME = ,E19.11./	PITE(6,207)DT = ,EI9.11) -crmat (22H STEP SIZE = ,E19.11) -crmat (22H STEP SIZE = ,E19.11)	IF(DT3. NE.0.) GO TC 11	RETE(6,206) TI,TF ARITE(6,208)DT.TI,TFL,DT2,TFL,TF ARITE(6,208)DT.TI,TFL,DT2,TFL,TF ARMAT (22H STEP_SIZE = ,E19.LL,13F BETWEEN T = ,E19.1L,	C TO 12 F=TF3 RITE(6,208) DT,TI,TF1,DT2,TF1,TF2,DT3,TF2,TF RAD(5,50C1) (C(1), T=1.8)	-ČRMÁT (8FI0.4) read (5,1c6)(x(1),1=1,NN) -Crmat (3F25.5)	J = 0 C 14 I=1,8 F(C(I).NE.0.) J=J+1 CATINUE	C 16 I = 1, NN F(X(I), NE+0.) K=K+1	LN INUE [F(J - 1)17,18,19 [RITE (6,2C9) [CRMAT (/,344 ALL THE CCNSTANTS, C(I), ARE ZERC)	RITE (6,210) RETE (6,210) CRMAT (1,30H THE CNLY NON-ZERC CCNSTANT IS)	RITE (6,211) FRMAT (7,35H THE NON-ZERO CONSTANTS, C(1), ARE) C 422 I=1,8	F(C(I) *NE *0 *) WRITE(5,212) I,C(I) ECRMAT (14X,2HC(,12,4H) = ,E12.4)	CENTINUE [F(K - 1)424,425,426 NRITE (6,1209) Strendt (7,364) // The Initian Feritions are 7ero 1	ECTO 20 RITE (6,1210)
50 C C	206	207	5	208	11	5001 106	14	2	1c 17 205	21C	5110 717 75	212	171120 171700 7770 17170	4 2 1

) Pred Pred (Pred Pred	hand hand hand hand hand ha		4 Jane 1994 1994		4 part part part	4 } { } { } { }				
<pre>0 FCRMAT (/,39H THE CNLY NON-ZERC INITIAL CCNCITION IS) 6 KRITE (6,1211) 1 FCRMAT (/,36H THE NON-ZERO INITIAL CONDITIONS ARE) 1 FCRMAT (/,36H THE NON-ZERO INITIAL CONDITIONS ARE) 1 F(X(I).NE.0C) WRITE(6,1212) I,X(I) 2 FCRMAT (14X,2HX(,12,4H) = ,E25.5) 2 FCRMAT (14X,2HX(,12,4H) = ,E25.5) 3 CCNTINUE 3</pre>	CHECK FCR THE NUMBER OF COLUMNS CALLED FOR BY LOCATING FIRST	I F(JTITLE(J).EQ.IELANK) GO TO 22 I CCNTINUE J = 9 2 J = J - 1 2 J = J - 1	JJ IS NCW THE NUMBER OF COLUMNS. REPEAT WITH THE GRAPHS.	<pre>REAC (5,105)(KTITLE(1),KTITLE(1+1),IG(1),IG(1+1),I=1,7,2) FCRMAT (4(2A8,2I2)) CC 24 K=1,7,2 IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII</pre>	IF(KTITLE(K).EG.IBLANK.AND.KTITLE(K+1).EG.IBLANK) GC TO 25 I 24 CCNTINUE 15 k = 8	C KK = K/2 KKK = KK*2 VI T T D = 0	IF(KK.NE.I) GO TO 3C6 IF(IG(3) + IG(4) EC.0) GO TO 306 IF(IG(5) + IG(6).NE.0) GO TO 306 N(1TIP = 2	KKK = 4 GC TO 366 15(IG(7) + IG(8).NE.0) GO TC 305		IF MULTIP = 0, KK IS THE NUMBER CF SINGLE CLRVE GRAPHS. CTHERMISE I MLLTIP IS THE NUMBER CF CURVES CN A SINGLE GRAPH.	:¢ IF(JJ.EG.0) GO TO 27
1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	ວ 	(1) (1)	3()(61	61		0 M	0 19	()()()()	0 0 0
	0000	Ç									

.

INT12850 INT12850 INT12850 INT12860 INT12870 INT12880 INT12880 INT12880 INT12800 INT12500

.



				-1	- j j j			- k k k k	- h h h h		- her her her		
	ZZZZZZ												
BLE		AF	U Z		AF		å						
۷Ī		ES	ГЦ		ES		00						
I A R		ABL	ND ND		18 L	[2	_						
5		R I /	XU XV		RIA	2	AIA						
NI		∧	•α~ ×α~		ΝA	ХТ ХХ	2						
CND		UZ			Ċ		ING						
SPI		C II	NUL+		C I I	10 10	цъ						
RE		NDa	++		ND	2	NT						
C C R		S P			ESF		ш						
ш		Na-			2 2	U L	2						
Ц Т	~	HOU		_	U U		ш						
D Z	D	SH-	110 111 111 111	~	щ	116	ய						
n N	A L	5	HT HT S	ED	F	X (ΙΖ						
L'S		+2×	44. 44.	H H	AND	· · ·	I⊿L						
	α.	S S S S S S	N U	с Ш	ш	<1>	LI						
-Д- -Ш-	IS	□-> 	Y LL	<u>م</u> د	1L	Ц Ц Ц	I N						
				A R E	1	12	• (5						
NUX C+	T01	YI.	24	s	Нd	XHO	1 N (
LL L		TA -		НЧ	ßA	-X-	EΡ						
ш Ц Ш С К		JULT Zm X		GR /	ш	П. 12 12	Ϋ́Η						54
HH a		NTH OHE	エエト	υ	TH	14X 741	У D		大大				D
			₹ [±] ,	Z	H	XØ.	80		ANA ANA	~			-
200		DANC C	2 ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	17	.5.	-2>	шН			С ж	Z~		60
-14	410	101-1	NIA	12	22	1 X X X	-			< • •0 •F	21	•	0
				-2-	172		V C S					11	е ш
Шчц		L L L L L	B⊢ DAAmc			ATC	ជ		າແມ່ນແມ່	2	-27-	NN NN	- -
L A A			-HZH-		121-1-2		- IS		L,]ran]ran]ran . _ pang peng peng . 			10	<u> </u>
in the second se				2 SILC	25120	131 131 131	F	H-22		x U U U		UR	IF
14	1210	.16	117	17	5.0	51					42		CC
10		^v V	(III V	h		~ ~	11.11	,					250



	<pre>INCPR = C(11)+0.000C0001 C(11) = 20. IF(MOD (NCPTS, 50*INCPR).EQ.0) GC TO 46 IF(MOD (NCPTS, 10*INCPR).EQ.0) GC TO 46 IF(MOD (NCPTS, 10*INCPR))54,48,54 IF(MOD (NCPTS, 10*INCPR))54,48,54 MRITE (6,218) (ITITLE(1),1=1,6),IPAGE,ITITLE(7),(JTITLE(1),1=1) MRITE (6,218) (ITITLE(1),1=1,6),IPAGE,ITITLE(7),(JTITLE(1),1=1)</pre>	<pre>GC 10 4/ 1047 hRITE (6,1218)(ITITLE(I),I=1,6),IPAGE,(JTITLE(I),I=1,8) hRITE (6,219) 47 hBITE(6,215) 218 FCRMAT (1H1.//.23X.6A8.7X.5HPAGE .12.14h CF CUTPUT FGR.A8./// 218 FCRMAT (1H1.//.23X.6A8.7X.5HPAGE .12.14h CF CUTPUT FGR.A8.///</pre>	1216 FCRMAT(1H1,///,23X,6A8,27X,5HPAGE,12,////,21X,4(A8,17X)) 215 FCRMAT(1H) 46 DC 45 I=1,NN 45 XC(1) = X(1) 7C = T 8 FTURN R FTURN	5C CC 53 I=1,JJ	<pre>PR(I) = T IF(IP(I).NE.0) PR(I)=XC(IP(I)) 52 CCNTINUE</pre>	22C FCRMAT (7X, 8E25.5) 54 IF(KK.EC.0) GG TO 62 INCGR = C(12)+0.C00001	57 XC(I) = X(I) TC = T C(10) = 4.	SECC 61 I=1,KKK	<pre>6 GR(I) = T IF(IG(I).NE.0) GR(I)=XC(IG(I)) CINTINUE CINTINUE KF1 = KKK.GE.8) GO TO 1610 KF1 = KKK + 1 CC 1612 I=KP1,8</pre>
	*	•			1				
	(I	Q Q							
	<u> </u>	•	×						
(I), A8, (X), X), X, Y,	ш	C A	~						
E(I), 7X)), 7X)),	1	° ∪ 4	·						
TLE(I), ,8) ,17X))	H		w						
ITLE(I), 1,8) FOR,A8, E,17X))	L L		V						
TTTLE(I), =1,8) =1,8) T FOR,A8, A6,17X))	3	, I P							
(JTITLE(I), ,I=1,8) PUT FOR,A8, 4(A8,17X))	•	~ +							
<pre>,(JTITLE(I),</pre>	~ ~		X						
<pre>/),(JTITLE(I), (I),I=1,8) CUTPUT FOR,A8, LX,4(A6,17X))</pre>	5	ш́ С	2						
(7),(JTITLE(I), E(I),I=1,8) CUTPUT FOR,A8, 21X,4(A6,17X))	ш.	UL.							
E(7),(JTITLE(1), LE(1),I=1,8) F CUTPUT FOR,A8, ,21X,4(A6,17X))	ب ب	LI O							
TLE(7),(JTITLE(1), ITLE(1),I=1,8) CF CUTPUT FOR,A8, //,21X,4(A6,17X))	i i i i i i i i i i i i i i i i i i i	+ 1							
ITLE(7),(JTITLE(1), TITLE(1),I=1,8) h CF CUTPUT FOR,A8, ///,21X,4(A6,17X))									
TITLE(7), (JTITLE(1), JTITLE(1), I=1,8) L4h CF CUTPUT FOR,A8, /////21X,4(A8,17X))	44 •	D D							
46 .ITITLE(7),(JTITLE(1), .(JTITLE(1),I=1,8) .14h CF CUTPUT FOR,A8, ./////,21X,4(A6,17X))		Ш2 Ц2	[2]				\sim		
<pre>2 46 2 47 3E.ITITLE(7).(JTITLE(1). 3E.(JTITLE(1).I=1.8) 12.14h CF CUTPUT FOR,A8. 12.1////.21X.4(A8.17X)) 52</pre>	A 11	A (~>		
TC 46 TC 47 AGE,ITITLE(7),(JTITLE(1), AGE,(JTITLE(1),I=1,8) ,12,144 CF CUTPUT FOR,A8, ,12,////,21X,4(A8,17X)) ,22,/////,21X,4(A8,17X))		<u>a</u>					2		
TC 47 TC 47 PAGE, ITITLE(7), (JTITLE(1), PAGE, (JTITLE(1), I=1,8) : 12,14h CF CUTPUT FOR, A8, : 12,////,21X,4(A8,17X)) : 12,/////,21X,4(A8,17X))		- U	5 5						
<pre>GC TO 46 GC TO 47 ,IPAGE,ITITLE(7),(JTITLE(1), ,IPAGE,(JTITLE(1),I=1,8) GE ,12,14h CF CUTPUT FOR,A8, GE ,12,/////,21X,4(A8,17X)) GE ,12,/////,21X,4(A8,17X)) TC 62</pre>			4				0		
<pre>GC TO 46 GC TO 46).IPAGE,ITITLE(7),(JTITLE(1),).IPAGE,(JTITLE(1),I=1,8) AGE ,12,14h CF CUTPUT FOR,A8, AGE ,12,1////21X,4(A8,ITX)) AGE ,12,/////21X,4(A8,ITX)) AGE TO 62</pre>	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	• H	L, L,		2		0		-
0) GC TO 46 48,54 6),IPAGE,ITITLE(7),(JTITLE(1), 6),IPAGE,(JTITLE(1),I=1,8) HPAGE,12,14h CF CUTPUT FOR,A8, HPAGE,12,1////,21X,4(A8,I7X)) HPAGE,12,/////,21X,4(A8,I7X)) GO TC 62 GO TC 62	0 0 0+ pmd	~ v	u u				2		H
<pre>0) GG TO 46 48.54 .1.6).IPAGE.ITITLE(7).(JTITLE(1). .1.6).IPAGE.(JTITLE(7).(JTITLE(1). .1.6).IPAGE.(JTTLE(1).1=1.8) SHPAGE .12.14h CF CUTPUT FOR.A8. SHPAGE .12.1/////21X.4(AG.17X)) SHPAGE .12.//////21X.4(AG.17X)) .5hPAGE .12./////21X.4(AG.17X)) .5hPAGE .12.////21X.4(AG.17X)) .5hPAGE .12.////21X.4(AG.17X)) .5hPAGE .12.////21X.4(AG.17X)) .5hPAGE .12.////21X.4(AG.17X)) .5hPAGE .12.////21X.4(AG.17X)) .5hPAGE .12.////21X.4(AG.17X)) .5hPAGE .12.////21X.4(AG.17X)) .5hPAGE .12.////21X.4(AG.17X)] .5hPAGE .12.////21X.4(AG.17X)] .5hPAGE .12.////21X.4(AG.17X)] .5hPAGE .12.////21X.4(AG.17X)] .5hPAGE .12.////21X.4(AG.17X)] .5hPAGE .12.////21X.4(AG.17X)] .5hPAGE .12.////21X.4(AG.17X)] .5hPAGE .12.////21X.4(AG.17X)] .5hPAGE .12.////21X.4(AG.17X)] .5hPAGE .12.///21X.4(AG.17X)] .5hPAGE .12.///21X] .5hPAGE .12.///21X] .5hPAGE .1</pre>	034 H		* * * *		Ĕ	^			S S
E9.0) GC TO 46 54,48,54 1=1.6),IPAGE,ITITLE(7),(JTITLE(1), 1=1.6),IPAGE,(JTITLE(1),1=1.8) X.5HPAGE,12,14H CF CUTPUT FOR,A8, X.5HPAGE,12,1////21X,4(A8,17X)) P(1)) P(1)) 0) GO TC 62 (1))	• • ~ *	~ ~	N				ш		ted.
<pre>-E9.0) GC TO 46 ;E9.0) GC TO 46 .I=1.6).IPAGE.ITITLE(7).(JTITLE(1). .I=1.6).IPAGE.(JTITLE(1).I=1.8) 7X.5HPAGE .I2.14H CF CUTPUT FOR,A8. 7X.5HPAGE .I2.14H CF CUTPUT FOR,A8. 7X.5HPAGE .I2.14H CF CUTPUT FOR,A8. 17X.5HPAGE .I2.1/////21X.4(A8.17X)) .IP(1)) IP(1)) IP(1)] iG(1)]</pre>	- aaa i	1)			C C	•• •••	2		000
I R):EQ.0) GC TC 47 R):EQ.0) GC TC 47 I).I=1.6).IPAGE.ITITLE(7).(JTITLE(1). I).I=1.6).IPAGE.(JTITLE(7).(JTITLE(1). B.7X.5HPAGE .12.14h CF CUTPUT FOR,A8. 27X.5HPAGE .12.14h CF CUTPUT FOR,A8. 27X.5HPAGE .12.1////21X.4(A6.17X)) .27X.5HPAGE .12./////21X.4(A6.17X)) I.JJ) I.JJ) C(IP(I)) C(IG(I)) C(IG(I))	0 222 ~~	- 4	~~a		×	H 0	~		× Ť
01 PR):EQ.0) GC TO 45 PR))54,48,54 7 7 7 7 7 1),1=1,6),1PAGE,1TITLE(7),(JTITLE(1), A8,7X,5HPAGE,12,14h CF CUTPUT FOR,A8, A8,7X,5HPAGE,12,14h CF CUTPUT FOR,A8, 9,27X,5HPAGE,12,1////21X,4(A8,17X)) 9,27X,5HPAGE,12,/////21X,4(A8,17X)) 1,0 01).NE.0) GO TC 62).NE.0) GO TC 62 10 XC(fiG(1)) 10 10	0 000 40	u v				H NO	<u>с</u>		16
0001 VCPR),EQ.0) GC TC 49 VCPR),54,48,54 CCPR),54,48,54 CCPR),54,48,54 CCPR),1=1,6),1PAGE,1TITLE(7),(JTITLE(1), E(1),1=1,6),1PAGE,(JTITLE(1),1=1,8) CC(1),1=1,6),1PAGE,12,144,CF CUTPUT FOR,A8, CA8,7X,5HPAGE,12,144,CF CUTPUT FOR,A8, COPR,27X,5HPAGE,12,144,CF CUTPUT FOR,A8, COPR,21X,5HPAGE,12,144,CF CUTPUT FOR,A8, COPR,21X,5HPAGE,21X,5H		F ×	· · ·		H	~~~00	ŭ		
<pre>GC001 INCPR):EQ.0) GC TO 46 INCPR))54,48,54 INCPR))54,48,54 INCPR))54,48,54 IO47 ILE(I),I=1,6),IPAGE,ITITLE(7),(JTITLE(I), TLE(I),I=1,6),IPAGE,(JTITLE(7),(JTITLE(I), TA,6A8,7X,5HPAGE,I2,14h CF CUTPUT FOR,A8, 7X,6A8,7X,5HPAGE,I2,14h CF CUTPUT FOR,A8, 7X,5A8,7X,5HPAGE,I2,14h CF CUTPUT FOR,A8, 7X,5A8,7X,5HPAGE,I2,14h CF CUTPUT FOR,A8, 7X,5A8,7X,5HPAGE,I2,14h CF CUTPUT FOR,A8, 7X,5A8,7X,5HPAGE,I2,14h CF CUTPUT FOR,A8, 7,1=1,JJ) 1)=XC(IP(I)) 1)=XC(IP(I)) 1)=XC(IP(I)) 1)=XC(IP(I)) 1)=XC(IP(I)) 1)=XC(IG(I)) 1)=XC(IG(I)) 1)=XC(IG(I))</pre>	0 ** 1	I m	X		~	HS O	2		~ 0
0666001 0*INCPR): EQ.00 GC TO 46 INCPR): EQ.00 GC TO 46 ITLE(1).1=1.6).IPAGE.ITITLE(7).(JTITLE(1). FITLE(1).1=1.6).IPAGE.(JTITLE(7).(JTITLE(1). 23X,6A8.7X,5HPAGE .12.14h CF CUTPUT FOR.A8. 23X,6A8.27X,5HPAGE .12.14h CF CUTPUT FOR.A8. 23X,6A8.27X,5HPAGE .12.1////21X,4(A8.17X)) 3,6A8.27X,5HPAGE .12.1////21X,4(A8.17X)) 1,1=1.JJ) 3,000001 1,1=1.JJ) 1,1=1.JJ]	• NH FH		-000/		d	RUF.	·		0
.0000001 50*INCPR):E0.00 GC TO 46 I0*INCPR):54.48.54 IIIILE(I).I=1.6).IPAGE.ITITLE(7).(JTITLE(I). IIIILE(I).I=1.6).IPAGE.(JTITLE(7).(JTITLE(I). ITITLE(I).I=1.6).IPAGE.IZ.144 CF CUTPUT FOR.A8. 331775) 31775 3177	0+ ~	~ ~	A.+			AN O			C ဘ
0.0000001 9:50*:INCPR):E9.0) GC T0 49 10*:INCPR):E9.0) GC T0 47 1 INCPR):E9.0) GC T0 47 (ITTLE(I),I=1,6),IPAGE,ITTLE(7),(JTTLE(I),I=1,8) (ITTLE(I),I=1,6),IPAGE,1Z,144 CF CUTPUT F0R,A8, 223X;6A8,7X;5HPAGE,1Z,144 CF CUTPUT F0R,A8, 223X;6A8,27X;5HPAGE,1Z,1////2IX,4(A8,17X)) PR(I)=XC(IP(I)) PR(I)=XC(IP(I)) PR(I)=XC(IP(I)) C0 000001 C0 000001 C0 000001 C1 0 1610 C1 0 1610		8	41~		6	~000~	F-	¥	
)+0.0000001 T5;50*INCPR],EQ.00 GC T0 46 T5;10*INCPR])54,48,54 T5;10*INCPR])54,48,54 T5;10*INCPR])54,48,54 GO T0 10471),1=1,6),1PAGE,1TITLE(1),1=1,8) 3)(ITTLE(1),1=1,6),1PAGE,12,144 CF CUTPUT FOR,A8, (//,233,6A8,7X,5HPAGE,12,144 CF CUTPUT FOR,A8, //,233,6A8,7X,5HPAGE,12,1////,21X,4(A8,17X)))) PR(I)=XC(IP(I)) 0) PR(I)=XC(IP(I)) 1) PR(- 0000 W-		n 40	2	•	N O	0.20	X	·
<pre>1)+0.0000001 [FTS;50*INCPR]:E9:0) GG TG 46 [FTS;10*INCPR])54,48,54 GE +1 INCPR])54,48,54 GE +1 II TTLE(I),I=1,6),IPAGE,ITTLE(I),I=1,8) S9 (TTTLE(I),I=1,6),IPAGE,(JTTLE(I),I=1,8) S9 (TTTLE(I),I=1,6),IPAGE,I2,144, CF CUTPUT FOR,A8, 7/7,233,6A8,7X,5HPAGE ,I2,144, CF CUTPUT FOR,A8, 7/7,7,233,6A8,77,5HPAGE ,I2,144, CF CUTPUT FOR,A8, 7/7,7,233,5A8,77,5HPAGE ,I2,144, CF CUTPUT FOR,A8, 7/7,7,7,233,5A8,77,5HPAGE ,I2,144, CF CUTPUT FOR,A8, 7/7,7,7,233,5A8,77,5HPAGE ,I2,144, CF CUTPUT FOR,A8, 7/7,7,7,233,5A8,77,5HPAGE ,I2,144, CF CUTPUT FOR,A8,77,5HPAGE ,I2,144, CF CUTPUT FOR,A8,77,77,77,77,77,77,77,77,77,77,77,77,77</pre>	-OZZZE-NH				z.	NXO	·Z ••• •	×	
<pre>(11)+0.0000001 NGPTS, D0*:NNCPR): EQ.00) GG TG 49 NGPTS, D0*:NNCPR)) 54,48,54 NGPTS, D0*:NNCPR)) 54,48,54 NGPTS, D0*:NNCPR)) 54,48,54 NGPTS, D1 1047 D1 104 D1 1047 D1 10</pre>	No HITTON	~ ~ N-	M XHHIN	-	H •	ar ou	n		H • UIXH
<pre>2(11)+0.060001 2(11)+0.060001 NCPTS; 10*INCPR): E9.0) EC TQ 46 NCPTS; 10*INCPR): E9.0) FAGE, 17TLE(T), 1=1,8) 2159 2159 2159 2159 2159 2159 2159 2159</pre>		4000		 				" 	
<pre> C(L1)+0.0000001 C(CPTS; 50*!NCPR): Eq.0) EC TQ 46 N(CPTS; 50*!NCPR): Eq.0) EC TQ 46 N(CPTS; 10*!NCPR): Eq.0) EC TQ 46 N(T] TLE(1), I=1,6), IPAGE, IJTILE(1), I=1,8) N(T) TLE(1), I=1,6), IPAGE, (JTITLE(1), I=1,8) N(T) TLE(1), I=1,6), IPAGE, (JZ), A(A8,17X)) N(T) TLE(1), I=1,6), IPAGE, IZ, ITLE(1), I=1,8) N(T) TLE(1), I=1,6), IPAGE, IZ, ITLE(1), I=1,8) N(T) TLE(1), I=1,0) TI=1,10 N(T) TLE(1), I=1,0) S(12) +0.000001 S(12) +0.0000001 S(12) +0.000001 S(12) +0.0000001 S(12) +0.000001 S(12) +0.000001 S(12) +0.000001 S(12) +0.000001 S(12) +0.0000001 S(12) +0.000001 S(12) +0.0000001 S(12) +0.000001 S(12) +0.000001 S(12) +0.0000001</pre>	000	⊂⊢	HH HZ		" SZ		"O" ⊢":	Z	-ZY -
<pre>2(11)+0.0000001 00 (NCFTS; 10*:NCPRN); 54,48,54 100 (NCFTS; 10*:NCPRN); 54,48,54 100 (NCFTS; 10*:NCPRN); 54,48,54 165,2169 (1717LE(1),1=1,6),1PAGE,1717LE(1),1=1,8) (6,2169) (1717LE(1),1=1,6),1PAGE,(JT17LE(1),1=1,8) (6,2159) (1717LE(1),1=1,6),1PAGE,12,144,CF CUTPUT FOR,A8, (6,2159) (1717LE(1),1=1,6),1PAGE,12,144,CF CUTPUT FOR,A8, (6,22159) (1717LE(1),1=1,0) (1717,10,1) (1717,10,1) (1717,10,1) (171,1</pre>				- m m	AL L	H Y Y Y Y		¥	
<pre>BR = 2(11)+0.0000001 MDD (NCPTS; 50* INCPR): 54,96 MDD (NCPTS; 10* INCPR): 54,96 FACE +1 RE (6:2) B(TTTLE(1),1=1,6), IPAGE, ITTLE(1), (JTTLE(1), FE (6:2) B(TTTLE(1),1=1,6), IPAGE, (JTTLE(1),1=1,8) FE (6:2) B(TTTLE(1),1=1,0) FE (6:2) P(T) FE (6:2) P(T) FE (6:2) P(T) FE (6:2) P(T) FE (6:2) P(T) FE (1) FE (1) F</pre>	OH-JACHH					HA CO.		_ `	
<pre>CFR = C(11)+0.00C001 MOD (NCFTS; 50*NNCFR); 59,90) GC TC 46 MOD (NCFTS; 10*INCFR); 54,46, 54 MOD (NCFTS; 10*INCFR); 54,46, 54 MOD (NCFTS; 10*INCFR); 54,46, 54 MOD (NCFTS; 10*INCFR); 54,46, 54 MOD (NTTLE(1),1=1,6),1PAGE,1TITLE(1),1=1,8) TF (4; 2; 2; 9) (1717LE(1),1=1,6),1PAGE,(JTTLE(1),1=1,8) TF (4; 2; 2; 9) (1717LE(1),1=1,6),1PAGE,12,144, CF CUTPUT FOR,AB, TF (4; 2; 2; 9) (1717LE(1),1=1,6),1PAGE,12,144,08,17X)) TF (1; 1; 1,7,1) TF (4; 2; 2; 9) (1717LE(1),1=1,9,1) TF (4; 2; 2; 9; 1] (1,1=1,9,1) TF (4; 2; 2; 9; 2] (1,1=1,9,1) TF (4; 2; 2]</pre>				0		CULZ'		μ <u>Ο</u>	CHUMACO CHUMACO
<pre>NNCPR = C(11)+0.000001 Het NoD (NCPTS; 50*NNCPR): Eq 46 Het NoD (NCPTS; 10*NNCPR): Eq 46 Het NoD (NCPTS; 10*NNCPR): Eq 49 Het NoD (NCPTS; 10*NNCPR): Eq 49 Het NoD (NCPTS; 10*NNCPR): Eq 49 Het NoD (NCPTS; 10*NNCPR): Eq 40 Het NoD (NCPTS; 10*NCPR): NE: 0) GO TC 62 Het NoD (NCPTS; 10*NCPR): NE: 0) GO TC 62</pre>			-		<u> </u>				لايا دينه الي
<pre>NCCR = C(11)+0.0000001 FF(MDD (NCFTS; 50*NNCPR); 54.48, 54 FF(MDD (NCFTS; 10*NNCPR); 54.48, 54 FFAND (NCFTS; 10*NNCPR); 54.48, 54 FFANT (66.216) (1711Le(1),1=1,6),1PAGE,1711Le(1),1=1,8) NRTE (66.218) (1711Le(1),1=1,6),1PAGE,12111Le(1),1=1,8) NRTE (66.218) (1711Le(1),1=1,6),1PAGE,12111Le(1),1=1,8) NRTE (66.218) (1711Le(1),1=1,6),1PAGE,12111Le(1),1=1,8) NRTE (66.218) (1711Le(1),1=1,6),1PAGE,12111Le(1),1=1,8) NRTE (66.218) (1711Le(1),1=1,6),1PAGE,12111Le(1),1=1,8) FCNAT (1H1)//2337,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)//2337,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)//2337,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)//223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)//223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)//223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)//223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)//223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12,1////21X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12,1////22X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12,1////22X,5HPAGE,12X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12,1////22X,5HPAGE,12X,4(A8,17X)) FCNAT (1H1)///223X,6A8,7X,5HPAGE,12X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11//2X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11///22X,11//2X,11//2X,11//2X,11//2X,11//2X,11//2X,11//2X,11//2X,11//2X,11//2X,11//2X,11//2X,11//2X,11//2X,11//2X,11//2X,11//2X,11/</pre>	40	7 7 7		20	(1) (1)	04	5	ω Ω'	ć 1
<pre>C C C C C C C C C C C C C C C C C C C</pre>		ON				N		- 1	-
<pre>NCPR = C(11)+0.0000001 If MOD (NCPTT; 50* INCPR): E0.0) GC T0 46 If MOD (NCPTT; 0* INCPR): E0.0) GC T0 46 If MOD (NCPTT; 0* INCPR): E0.0) ICC T0 46 NETTE (0.218) (TT1TE(1), 1=1,6), IPAGE, ITTTLE(1), (JTTLE(1), NETTE (0.218) (TT1TLE(1), 1=1,6), IPAGE, (JTTLE(1), 1=1,8) NETTE (0.212) NETTE (0.212) NETTE (0.212) 218 FCRMAT (1AH), (/233X; 6A8, 7X; 5HPAGE , 12, ////, 21X; 4(A8, 18) 218 FCRMAT (1AH), (/233X; 6A8, 27X; 5HPAGE , 12, ////, 21X; 4(A8, 17X)) 218 FCRMAT (1AH), (/233X; 6A8, 27X; 5HPAGE , 12, ////, 21X; 4(A8, 17X)) 218 FCRMAT (1AH), (/233X; 6A8, 27X; 5HPAGE , 12, ////, 21X; 4(A8, 17X)) 218 FCRMAT (1AH), (/233X; 6A8, 27X; 5HPAGE , 12, ////, 21X; 4(A8, 17X)) 218 FCRMAT (1AH), (/233X; 6A8, 27X; 5HPAGE , 12, ////, 21X; 4(A8, 17X)) 228 FCRMAT (1AH), (/233X; 6A8, 27X; 5HPAGE , 12, ////, 21X; 4(A8, 17X)) 238 FCRMAT (1AH), (/220) PR(1) = 3. 706 (17 T = X(1)) 238 FCRMAT (1AH), (/220) PR(1) = 1.JJJ) 238 FCRMAT (1AH) = 7 706 (17 T = X(1)) 238 FCRMAT (1AH), (/233X; 6A8, 27X; 5HPAGE , 12, ////, 21X; 4(A8, 17X)) 238 FCRMAT (1AH), (/233X; 6A8, 27X; 5HPAGE , 12, ////, 21X; 4(A8, 17X)) 238 FCRMAT (1AH), (/233X; 6A8, 27X; 5HPAGE , 12, ////, 21X; 4(A8, 17X)) 238 FCRMAT (1AH), (/233X; 6A8, 27X; 5HPAGE , 12, ////, 21X; 4(A8, 17X)) 238 FCRMAT (1AH), (/232) (PR(1) = 1.JJJ) 248 FCRMAT (1AH), (/232) (PR(1), 1=1.JJJ) 258 FCRMAT (1AH), (/272) (PR(1), 1=1.JJJ) 278 FCRMAT</pre>		-			,			12.1	,

8)

~

~

٠

I T N N T I T N N T I T N N N T I T N N N T I T N N N T I T 1 1 4 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	INT14360	1NT14420	INT14440	INT14480 INT14480	INT14510 INT14520	INT14540	111145500000000000000000000000000000000	INT14610 INT14620		INT14680	INT14700	INT14730	INTI4750
G G R (1) C C R (1)	2 NCPTS = NOPTS = GK(8) 2 NCPTS = NOPTS + 1 1 F(NUMPTS.LT.1000) GO TG 64 1 D TS - 231 (000) GO TG 64	I FCRMAT (/////,26H STOP AT 1000 GRAPH PCINTS) GC TO 91	4 JŘ(NÖPÝŠ.LT.10000) GO TC 66 MRITE (6,222) 2 FCRMAT (/////,32H STOP AT 1000C INTEGRATICN STEPS)	6 IF(IPAGE - 15) 69,67,68 7 IF(MOD (NOPTS , 50*INCPR).NE.0) GC TO 65 8 MRITE (6,223) 379 5700 AT 10 0ACES FE OUTDIT	C TO 91 C TO 91 S C TO 151 S C TO 51 S		I KRITE (6,224) 4 FCRMAT (//////76H STOP WITH THE ABSCLUTE VALUE CF A DEPENDENT VARI 11ABLE GREATER THAN 1.0E+18. //57F INTEGRATICN PROEABLY UNSTABLE.I 2 TRY A SMALLER STEP SIZE. ,26HNG GRAPHS WILL BE PLCTTEC.)	2 01 = 0 (13) 2 01 = 0 (13) 3 16(11.1.1) 60 10 80 3 16(11.1.1) 60 10 75	4 WRITE (C,225) 5 FORMAT (/////,26H NORMAL STCP AT FINAL TIME) 1 FORMAT (/////,26H NORMAL STCP AT FINAL TIME)	5 IF(T.GE.TF1) GO TO 77	7 IF(T.GE(TF2) GO TO 79 8 C(12) = DT2	5 C(13) = DT3 6 TO 87	C ĬĔ(ŤĔ・ĞĖ・T) GO TC 74
161	Ŷ	22	22	2000	4	7(227		22	22		7	ŝ

,



PLOT 9 NULTIPLE £ 0 S ---ш SCAL 44 - + + $\times >$ CURVES Ē (NN, T, X, DT, C, TC, XC, CX) (V) mm 0 ×≻mm INDIVIDUAL ----×>-AXES S +0) , NUMPTS,01 (X2,Y2,NLMPTS,0 2 5 RUMPT BLE ALCNG RETURN 113411 13.312 13.312 -LT.T) 60 TC 76 = 5. 1000 4 PH-14 IF(C(10).EC.6.) R T = T + DT GC T0 2000 IF(KK.EC.0) G0 TC IF(WULTIP.NE.0) G NLWPTS=-NUMPTS DC 210 11.1.1.4 MRITE(6,99958) FCRNAT(1H1) 1711LE(5,99958) GC 10 (311,512,3 CALL PLCTP(X1,11 CALL PLCTP(X1,11 CALL PLCTP(X2,42) GC 10 310 (X2,42) CALL PLCTP(X3,33) GC 10 310 (X2,42) CALL PLCTP(X3,43) GC 10 310 (X2,42) CALL PLCTP(X4,42) CALL PLCTP(X4 CURVE 0 $m \neg \omega \neg$ (X3,Y3 41-N 1-UP RKUTTA YMMUD E C • • Р 0000 -12 F(TF) F(TF2) (10) INT ALL .5 ¢ Ц -----0 C HE n. 400 ~ ω ω 11 17 (n)~ C \mathbf{C} ω Ś S ----Ś 8 S 1-15 \sim Ū m (L) m mmin G ō ō -----C U U 000



	~	
<pre>TX(1) = 0. TX(2) = 0. TX(4) = SMLX TY(1) = SMLX TY(2) = BIGX TY(2) = SMLX TY(2) = SMLX TY(2</pre>	<pre>33C IF(NRC.NE.NR) GO TC ICOO IF(NR.GT.1) GO TC 333 ARITE(6,226) 226 FCRWAT(//,43H THE CNE RUN CALLEC FCR HAS BEEN CCMPLETED. ,//) 325 haite(6,227)NR 227 FCRMAT(//,5H THE ,11,37H RUNS CALLED FOR HAVE BEEN CCMPLETED.,// 57CP 57CP 57CP</pre>	Steroutine RKUTTA(/nn/,/T/,/X/,/ET/,/C/,/TC/,/XC/,/EX/) DIMENSICN X(30), C(15), XC(30), EX(30), C1(4), AK(4,30) REAL*8 AK,CT INCIC = C(10) - 4.04C.c0000001 If(INCIC.6T.1) G0 T0 3 C1(1) = 0.000 C1(2) = C.500 C1(2) = 0.500

00000000 **56665 56765 56765** លាលសាលាលាលាលា

52

.

.

C
<pre>(4, J)) / 6.0</pre>	*	CESCRIBEC UNCER SLERCUTINE ACCEPT LARGER PARAMETER INTI2550 FOR THE NEW FORMAT.		1 1.0	CC54200.CCC01C000.0C00 CC180000C00C0CC000C.0000 STATE 3
+ 7 + 7	'n	A P A		00.	
) AK(3,J)	*	A CARDS CLIFIED NT12360			00 • 0340 00 0000 3 6R CU
-l, J +(j, t)	÷	AR DAT			
AK(II- *(AK()	*	NS TH		• 00	000-01(200000 STAT
۰۲ ۲ ۲ ۲ ۲ ۲ ۲ ۲ ۲ ۲ ۲ ۲ ۲ ۲ ۲ ۲ ۲ ۲ ۲	- 1 7-	T A LA L	IS		00000000000000000000000000000000000000
)*01 C1(C1(1,1)		E MO MU NN NN NN	ALYS	10	1800 0000 TRI TRI
	¥	ARDS STAT	R AN	0000	1000 1000 1000
	₩ □ *		LASE	<i></i>	П105 000 000 000 000
		CWIN SAGE	СYЕ		50000 50000 51AT
	*		ANIC	0001	6 LET
	3 2000	モート	L L C C	100	S 1001

•

BIBLIOGRAPHY

- Sorokin, P.P. and Stevenson, M.J., "Stimulated Infrared Emission from Trivalent Uranium," <u>Physical Review Letters</u>, v. 5, no. 12, pp. 557-559, 15 December 1960.
- Brock, E.G., Csavinszky, P., Hormats, E., Nedderman, H.C., Stirpe, D., and Unterleitner, F., "Coherent Stimulated Emission from Organic Molecular Crystals," <u>Journal of Chemical Physics</u>, v. 35, no. 2, pp. 759-760, August 1961.
- Morantz, D.J., White, B.G. and Wright, A.J.C., "Stimulated Light Emission by Optical Pumping and by Energy Transfer in Organic Molecules," <u>Physical Review Letters</u>, v. 8, no. 1, pp. 23-25, 1 January 1962.
- Lempicki, A. and Samuelson, H., "Stimulated Processes in Organic Compounds," <u>Applied Physics Letters</u>, v. 2, no. 8, pp. 159-161, 15 April 1963.
- Sorokin, P.P. and Lankard, J.R., "Stimulated Emission Observed from an Organic Dye, Chloro-aluminum Phthalocyanine," <u>IBM Journal of Research and Development</u>," v. 10, pp. 162-163, March 1966.
- Sorokin, P.P. and Lankard, J.R., "Flashlamp Excitation of Organic Dye Lasers: A Short Communication," <u>IBM Journal of Research and</u> <u>Development</u>, v. 11, p. 148, March 1967.
- Schafer, F.P., Schmidt, W. and Volze, J., "Organic Dye Solution Laser," <u>Applied Physics Letters</u>, v. 9, no. 8, pp. 306-309, 15 October 1966.
- 8. Snavely, B.B., "Flashlamp-Excited Organic Dye Lasers," Proceedings of the IEEE, v. 57, no. 8, pp. 1374-1390, August 1969.
- Snavely, B.B. and Schafer, F.P., "Feasibility of CW Operation of Dye Lasers," <u>Physics Letters</u>, v. 28A, no. 11, pp. 728-729, 10 March 1969.
- Peterson, O.G., Tuccio, S.A. and Snavely, B.B., "CW Operation of an Organic Dye Solution Laser," <u>Applied Physics Letters</u>, v. 17, no. 6, pp. 245-247, 15 September 1970.
- Hercher, M. and Pike, H.A., "Continuous Dye Laser Emission from 5220 to 6570 Angstroms," <u>IEEE Journal of Quantum Electronics</u>, v. QE-7, no. 9, p. 473, September 1971.
- 12. Hercher, M. and Pike, H.A., "Tuneable Dye Laser Configurations," Optics Communications, v. 3, no. 1, pp. 65-67, March 1971.
- Hercher, M. and Pike, H.A., "Single Mode Operation of a Continuous Tuneable Dye Laser," <u>Optics Communications</u>, v. 3, no. 5, pp. 346-348, July 1971.

54

- 14. Kohn, R.L., Shank, C.V., Ippen, E.P. and Dienes, A., "An Intracavity-Pumped CW Dye Laser," <u>Optics Communications</u>, v. 3, no. 3, pp. 177-178, May 1971.
- Dienes, A., Ippen, E.P. and Shank, C.V., "High-Efficiency Tuneable CW Dye Laser," <u>IEEE Journal of Quantum Electronics</u>, v. QE-8, no. 3, p. 388, March 1972.
- 16. Tuccio, S.A. and Strome, F.C., "Design and Operation of a Tuneable Continuous Dye Laser," <u>Applied Optics</u>, v. 11, no. 1, pp. 64-73, January 1972.
- 17. Strome, F.C. and Tuccio, S.A., "Loss Analysis and Design Improvement for a Continuous Dye Laser," <u>IEEE Journal of Quantum Electronics</u>, v. QE-9, no. 2, pp. 230-235, February 1973.
- Shank, C.V., Bjorkholm, J.E. and Kogelnik, H., "Tunable Distributed-Feedback Dye Laser," Applied Physics Letters, v. 18, no. 9, pp. 395-396, 1 May 1971.
- 19. Chandra, S., Takeuchi, N. and Hartmann, S.R., "Prism-dye Laser," Applied Physics Letters, v. 21, no. 4, pp. 144-146, 15 August 1972.
- 20. Ippen, E.P. and Shank, C.V., "Evanescent-Field-Pumped Dye Laser," Applied Physics Letters, v. 21, no. 7, pp. 301-302, 1 October 1972.
- 21. Bloom, Arnold M., "CW Pumped Dye Lasers," <u>Coherent Radiation Tech-</u> nical Bulletin, 1972.
- 22. Sorokin, P.P., Lankard, J.R., Moruzzi, V.L. and Hammond, E.C., "Flashlamp-Pumped Organic-Dye Lasers," <u>The Journal of Chemical</u> Physics, v. 48, no. 10, pp. 4726-4741, 15 May 1968.
- Dewey, C.F. and Hocker, L.O., "Infrared Difference-Frequency Generation Using a Tunable Dye Laser," <u>Applied Physics Letters</u>, v. 18, no. 2, pp. 58-60, 15 January 1971.
- 24. Bradley, D.J., Nicholas, J.V. and Shaw, J.R.D., "Megawatt Tunable Second Harmonic and Sum Frequency Generation at 280 Nanometers from a Dye Laser," <u>Applied Physics Letters</u>, v. 19, no. 6, pp. 172-173, 15 September 1971.
- Hansch, T.W., "Repetitively Pulsed Tunable Dye Laser for High Resolution Spectroscopy," <u>Applied Optics</u>, v. 11, no. 4, pp. 895-898, April 1972.
- Baardsen, E.L. and Terhunc, R.W., "Detection of OH in the Atmosphere Using a Dye Laser," <u>Applied Physics Letters</u>, v. 21, no. 5, pp. 209-211, 1 September 1972.
- 27. Beiser, A., Perspectives of Modern Physics, McGraw-Hill, 1969.
- 28. Lower, S.K. and El-Sayed, M.A., "The Triplet State and Molecular Electronic Processes in Organic Molecules," <u>Chemical Review</u>, v. 66, no. 2, pp. 199-241, April 1966.



- 29. Condon, E.U., "The Franck-Condon Principle and Related Topics," <u>American Journal of Physics</u>, v. 15, no. 5, pp. 365-374, September-October 1947.
- 30. Kasha, M., "Characterization of Electronic Transitions in Complex Molecules," <u>Discussions of the Faraday Society</u>, no. 9, pp. 14-19, 1950.
- 31. Snavely, B.B. and Peterson, O.G., "Experimental Measurement of the Critical Population for the Dye Solution Laser," <u>IEEE Journal of</u> <u>Quantum Electronics</u>, v. QE-9, no. 10, pp. 540-545, October 1968.
- 32. Yariv, A. and Gordon, J.P., "The Laser," Proceedings of the IEEE, v. 51, no. 1, pp. 4-29, January 1963.
- 33. McColgin, W.C., Webb, J.P., Eberly, J.H. and Peterson, O.G., "Organic Dye Laser Threshhold," Journal of Applied Physics, v. 42, no. 5, pp. 1917-1928, April 1971.
- 34. Atkinson, J.B. and Pace, F.P., "The Spectral Linewidth of a Flashlamp-Pumped Dye Laser," <u>IEEE Journal of Quantum Electronics</u>, v. QE-9, no. 6, pp. 569-574, June 1973.
- 35. Maydan, D., "Fast Modulator for Extraction of Internal Laser Power," Journal of Applied Physics, v. 41, no. 4, pp. 1552-1559, 15 March 1970.
- 36. Sorokin, P.P., Lankard, J.R., Hammond, E.C. and Moruzzi, V.L., "Laser-Pumped Stimulated Emission from Organic Dyes: Experimental Studies and Analytical Comparisons," <u>IBM Journal of Research and Develop-</u> ment, v. 11, pp. 130-147, March 1967.
- 37. Bass, M., Deutsch, T.F., and Weber, M.J., "Frequency-and Time-Dependent Gain Characteristics of Laser-and Flashlamp-Pumped Dye Solution Lasers," <u>Applied Physics Letters</u>, v. 13, no. 4, pp. 120-124, 15 August 1968.
- 38. Bass, M. and Weber, M.J., "Frequency-and Time-Dependent Gain Characteristics of Dye Lasers," <u>IEEE Journal of Quantum Electronics</u>, v. QE-5, no. 4, pp. 175-187, April 1969.
- 39. Keller, Richard A., "Effect of Quenching of Molecular Triplet States in Organic Dyc Lasers," <u>IEEE Journal of Quantum Electronics</u>, v. QE-6, no. 7, pp. 411-416, July 1970.
- 40. Pappalardo, R., Samuelson, H., and Lempicki, A., "Calculated Efficiency of Dye Lasers as a Function of Pump Parameters and Triplet Lifetime," Journal of Applied Physics, v. 43, no. 9, pp. 3776-3787, September 1972.
- 41. Streifer, W. and Salt, P., "Transient Analysis of an Electronically Tunable Dye Laser-Part II: Analytic Study," <u>IEEE Journal of</u> of Quantum Electronics, v. QE-9, no. 6, pp. 563-569, June 1973.



- 42. Buettner, A.V., Snavely, B.B. and Peterson, O.G. in <u>Molecular</u> <u>Luminescence</u>, edited by E.C. Lim, p. 403, Benjamin, New York, 1969.
- 43. Mack, M.E., "Measurement of Nanosecond Fluorescence Decay Times," Journal of Applied Physics, v. 39, pp. 2483-2485, April 1968.
- 44. Bass, M. and Steinfeld, J.I., "Wavelength Dependent Time Development of the Intensity of Dye Solution Lasers," <u>IEEE Journal of</u> Quantum Electronics, v. QE-4, no. 2, pp. 53-58, February 1968.
- 45. Private Communication with Spectra Physics in February 1973.
- 46. Jacobs, R.R., Samuelson, H. and Lempicki, A., "Losses in CW Dye Lasers," Journal of Applied Physics, v. 44, no. 1, pp. 263-272, January 1973.

,

INITIAL DISTRIBUTION LIST

		No.	Copies
1.	Defense Documentation Center Cameron Station Alexandria, Virginia 22314		2
2.	Library, Code 0212 Naval Postgraduate School Monterey, California 93940		2
3.	Asst. Professor J. P. Powers, Code 52 Po Department of Electrical Engineering Naval Postgraduate School Monterey, California 93940		1
4.	LCDR Gerald W. Snyder, USN 1937 E. Ivy Avenue Saint Paul, Minnesota 55119		1
5.	Dr. Richard S. Hughes, Code 6043 Naval Weapons Center China Lake, California 93555		1
6.	Mr. Jim Jernigan, Code 6043 Naval Weapons Center China Lake, California 93555		1



SECURITY CLASSIFICATION OF THIS PAGE (When Data Entered)

REPORT DOCUMENTATION	PAGE	READ INSTRUCTIONS BEFORE COMPLETING FORM				
1. REPORT NUMBER	2. GOVT ACCESSION NO.	3. RECIPIENT'S CATALOG NUMBER				
4. TITLE (and Subtitie)	· · · · · · · · · · · · · · · · · · ·	5. TYPE OF REPORT & PERIOD COVERED				
The Wavelength Spectrum Shift of a Cavity-Dumped Argon Laser-Pumped Rhodamine 6-G Organic-Dye		Engineer's Thesis; September				
		1973				
		6. PERFORMING ORG. REPORT NUMBER				
7. AUTHOR(s)		8. CONTRACT OF GRANT NUMPER(a)				
Gerald Warner Snyder						
9. PERFORMING ORGANIZATION NAME AND ADDRESS		10. PROGRAM ELEMENT, PROJECT, TASK AREA & WORK UNIT NUMBERS				
Naval Postgraduate School						
Monterey, California 93940						
Naval Post graduate School		September 1973				
Monterey California 930/0		13. NUMBER OF PAGES				
Monterey, Carronnia 93940		60				
14. MONITORING AGENCY NAME & ADDRESS(II differen	t from Controlling Office)	15. SECURITY CLASS. (oi this report)				
Naval Postgraduate School		Unclassified				
Monterey, California 93940						
		15a, DECLASSIFICATION/DOWNGRADING SCHEDULE				
16. DISTRIBUTION STATEMENT (of this Report)						
Approved for public release; disti	ribution unlimit	ed				
17. DISTRIBUTION STATEMENT (of the abstract entered .	in Block 20, 11 different fro	em Report)				
18. SUPPLEMENTARY NOTES						
19. KEY WORDS (Continue on reverse side if necessary en	d identify by block number))				
Dye Laser Organic dye						
Dve Laser Rate Equations	Due Lason Rate Fountiona Pholonic ()					
Dve Laser Gain Equation						
-Jo Tabler darm Equation						
20. ABSTRACT (Continue on revorse elde if necessery and	i identify by block number)					
The wavelength spectrum of a CW pumped Rhodamine 6-G organic dye laser is						
pumped by 20 mass pulses at a super-bastoms toward shorter wavelengths when						
dumped by joinsee purses at a one meganeruz repetition rate from a cavity-						
and rate equations are developed for a simplified dve laser model and theo-						
rctical results are obtained for a computer simulation of the experiment.						
comparison is made of the theoretical and experimental results and satisfactor						
	and a start of the					
DD FORM 1472 EDITION OF LUCY OF IS OFFIC	CTE					
(Page 1) S/N 0102-014-6601						
	CECUDITY CLA	SSIFICATION OF THIS PAGE (When Data Enlaced				

.



20. (cont'd)

agreement is obtained within the limits of the values of the parameters used and the assumptions made in formulating the model.





